Author's response to reviews

Title: The use of complementary medicine by a self-selected community group of Australian women with polycystic ovary syndrome

Authors:

Susan Arentz Ms (s.arentz@uws.edu.au)
Caroline A Smith Professor (caroline.smith@uws.edu.au)
Jason A Abbott A/Professor (j.abbott@unsw.edu.au)
Alan Bensoussan Professor (a.bensoussan@uws.edu.au)

Version: 2 Date: 30 July 2014

Author's response to reviews: see over
30 July 2014

BMC Journal of Women’s Health,

Dear Madam/Sir,

Please find attached our manuscript titled ‘The use of complementary medicine by a self-selected community group of Australian women with polycystic ovary syndrome.’ MS: 1595428658137440. Please find the following revisions.

1. Abstract: Word count 327, sub-heading ‘Background’ inserted and acronym for complementary medicine (CM) removed.
2. Manuscript. Tables are inserted into the Table section following the references.
3. The reference list has a heading.

Thank you.

Yours sincerely

Susan Arentz